Author's response to reviews

Title: Growth of human gastric cancer cells in nude mice is delayed by a ketogenic diet supplemented with omega-3 fatty acids and medium-chain triglycerides

Authors:

Christoph Otto (otto_c@chirurgie.uni-wuerzburg.de)
Ulrike Kaemmerer (u.kaemmerer@mail.uni-wuerzburg.de)
Bertram Illert (illert_b@chirurgie.uni-wuerzburg.de)
Bettina Muehling (muehling_b@chirurgie.uni-wuerzburg.de)
Nadja Pfetzer (pfetzer_n@klinik.uni-wuerzburg.de)
Rainer Wittig (wittig@tavartis.com)
Hans Ullrich Voelker (hans-ullrich.voelker@mail.uni-wuerzburg.de)
Arnulf Thiede (thiede_a@chirurgie.uni-wuerzburg.de)
Johannes F Coy (coy@tavartis.de)

Version: 5 Date: 29 April 2008

Author’s response to reviews: see over
Cover letter with a point-by-point description of the changes made (proof reading)

Following changes was made:

1. Author list - the order was changed in the submission system and the authors names are written identically as they are in the submission system
2 ABSTRACT -> Abstract
3 Diets and Feedings -> Diets and feedings
4 author’s contributions -> authors’ contributions
5 nonparametric ranc variance test -> nonparametric rank order test (page 13)
6 References: a further reference was added: Langbein S et al. (ref 38)